Author's response to reviews

Title: Lifestyle interventions for overweight and obese pregnant women to improve pregnancy outcome: systematic review and meta-analysis

Authors:

Eugene Oteng-Ntim (eugene.oteng-ntim@gstt.nhs.uk)
Rajesh Varma (rajesh.varma@gstt.nhs.uk)
Helen Crocker (helen.crocker@ucl.ac.uk)
Lucilla Poston (lucilla.poston@kcl.ac.uk)
Pat Doyle (pat.doyle@lshtm.ac.uk)

Version: 6 Date: 27 April 2012

Author's response to reviews:

I have now loaded each figure and added the tables to the manuscript, but this have been extremely difficult on my computer and wondered whether you may have a more sophisticated system to do this effectively and timely. I have made enormous efforts though.